Author's response to reviews

Title: Evaluating a selective prevention program for binge drinking among young adolescents: Study protocol of a randomized controlled trial

Authors:

Jeroen Lammers (jlammers@trimbos.nl)
Ferry Goossens (fgoossens@trimbos.nl)
Suzanne Lokman (slokman@trimbos.nl)
Karin Monshouwer (kmonshouwer@trimbos.nl)
Lex Lemmers (llemmers@trimbos.nl)
Patricia Conrod (patricia.conrod@kcl.ac.uk)
Reinout Wiers (r.w.wiers@gmail.com)
Rutger Engels (r.engels@pwo.ru.nl)
Marloes Kleinjan (m.kleinjan@pwo.ru.nl)

Version: 3 Date: 17 February 2011

Author's response to reviews: see over
February 171th, 2011

Dear Sir / Madam,

I am enclosing a revised manuscript for submission to *BMC Public Health* entitled, “Evaluating a selective prevention program for binge drinking among young adolescents: Study protocol of a randomized controlled trial”. The total manuscript, including the title page, abstract, 1 tables, and 1 figure is 26 pages long. Co-authors are Drs. Ferry Goossens (Trimbos Institute), Drs. Suzanne Lokman (Trimbos Institute), Dr. Karin Monshouwer (Trimbos Institute), Dr. Lex Lemmers (Trimbos Institute), Dr. Patricia Conrod (University of London), Prof. Dr. Reinout Wiers (University of Amsterdam), Prof. Dr. Rutger Engels (Radboud University Nijmegen) and Dr. Marloes Kleinjan (Radboud University Nijmegen).

In the revised manuscript the following formatting changes have been made:
- Reference 44 and 45 are placed in the main text;
- All highlighting are removed from the manuscript;
- The figure at the end of the manuscript is cropped as much as possible to minimize white space around the figure.

I hope that the format of the manuscript meets the journal’s requirements and we look forward receiving your response.

Sincerely,

Jeroen Lammers

Jeroen Lammers (corresponding author)
Trimbos Institute
P.O. Box 725
3500 AS Utrecht The Netherlands
Phone: 0031-30-2959309
Fax: 0031-30-2971111
E-mail: jammers@trimbos.nl.